Author's response to reviews

Title: Serum tryptase, mast cells positive to tryptase and microvascular density evaluation in early breast cancer patients: possible translational significance

Authors:

Ilaria MARECH (imarech@ibero.it)
Michele AMMENDOLA (Mammedola@gmail.com)
Rosario SACCO (rsacco@alice.it)
Gennaro S CAPRIUOLO (gcapriulo@gmail.com)
Rosa PATRUNO (rpatruno@ibero.it)
Rosangela RUBINI (rrubini@ibero.it)
Maria LUPOSELLA (mluposella@lice.it)
Valeria ZUCCALA’ (vzuccala@gmail.com)
Eufemia SAVINO (esavino@libero.it)
Cosmo D GADALETA (cgadaleta@libero.it)
Domenico RIBATTI (domenico.ribatti@uniba.it)
Girolamo RANIERI (giroman@tiscali.it)

Version: 2 Date: 30 January 2014

Author's response to reviews:

Dear Ms. BATTAD, Accordingly to your suggestions, I have modified the ABSTRACT, I have included an AUTHOR's CONTRIBUTION SECTION and an ACKNOWLEDGEMENTS section.

Sincerely, Domenico RIBATTI